Author's response to reviews

Title: Analysis of surgical and oncological Outcome in internal and external Hemipelvectomy in 34 Patients above the Age of 65 Years at a mean Follow-up of 56 Months

Authors:

Wiebke K Guder (wiebke.guder@ukmuenster.de)
Jendrik Hardes (jendrik.hardes@ukmuenster.de)
Georg Gosheger (georg.gosheger@ukmuenster.de)
Marcel-Philipp Henrichs (marcel-philipp.henrichs@ukmuenster.de)
Markus Nottrott (markus.nottrott@ukmuenster.de)
Arne Streitbürger (streitb@ukmuenster.de)

Version: 7 Date: 31 January 2015

Author's response to reviews:

Dear Ladies and Gentlemen,

we are glad to hear that our manuscript has been accepted.

As requested, we clearly state in the Methods Section that in Germany ethical approval is not deemed necessary for this kind of study.

To facilitate the payment process, I'll attach the billing address and our VAT number with this letter:

Wiebke Guder
Open Access Publikationsfonds
Krummer Timpen 3
48143 Muenster
Germany

VAT Number DE126118759.

Sincerely,
Wiebke Guder